Author’s response to reviews

Title: Psychiatric disorder in early adulthood and risk of premature mortality in the 1946 British Birth Cohort

Authors:

MAX HENDERSON (max.j.henderson@kcl.ac.uk)
MATTHEW HOTOPF (matthew.hotopf@kcl.ac.uk)
IMRAN SHAH (i.shah@nshd.mrc.ac.uk)
RICHARD D HAYES (richard.hayes@kcl.ac.uk)
DIANA KUH (d.kuh@nshd.mrc.ac.uk)

Version: 3 Date: 25 February 2011

Author’s response to reviews:

Dear BMC editorial team

i have made all the amendments you asked for.

bw